Author's response to reviews

Title: Life Threatening Biopsy from Haemophilic IleoPsoas Pseudotumour

Authors:

Azan S Al Saadi (azanalsaadi@gmail.com)
Ali H Al Wadan (amresam@hotmail.com)
Samir A EL Hamarneh (hamarneh@hotmail.com)
Mohamed E Emad (mohammede77@hotmail.com)

Version: 4 Date: 24 June 2007

Author's response to reviews:

an Acknowledgements section was inserted which states that written informed consent was granted by the patient for publication.
2. spaces were removed